Author's response to reviews

Title: Lifestyle factors, demographics and medications associated with depression risk in an international sample of people with multiple sclerosis

Authors:

Keryn L Taylor (keryn.taylor@svha.org.au)
Emily J Hadgkiss (emily.hadgkiss@hotmail.com)
George A Jelinek (george.jelinek@svha.org.au)
Tracey J Weiland (tracey.weiland@svha.org.au)
Naresh G Pereira (naresh.pereira@gmail.com)
Claudia H Marck (claudia.marck@svha.org.au)
Dania M van der Meer (dania.vandermeer@svha.org.au)

Version: 2 Date: 2 August 2014

Author's response to reviews:

We have added in line numbers and a statement in the methods about the consent.

Kind regards